Author's response to reviews

Title: Hemodynamic monitoring using a single-use indwelling transesophageal echocardiography probe in an unstable patient after open-heart surgery

Authors:

Emmanuelle Begot (emmanuellebegot@yahoo.fr)
Marc Clavel (marc.clavel@chu-limoges.fr)
Alessandro Piccardo (alessandro.piccardo@chu-limoges.fr)
Remy Bellier (remybellier@yahoo.fr)
Bruno Francois (b.francois@unilim.fr)
Nicolas Pichon (nicolas.pichon@chu-limoges.fr)
Philippe Vignon (philippe.vignon@unilim.fr)

Version: 3
Date: 22 June 2015

Author's response to reviews: see over
To the Editor,

We thank you for accepting our manuscript for publication in *BMC Medical Imaging*. As requested, we modified our manuscript according to the Executive Editor’s comments:

1. We added a Consent section at the end of our manuscript stating that we obtained the consent of the next of kin of the patient since the patient died in the hospital. (see page 7)

2. We edited our manuscript to improve the style of the language. The changes appear in color. We hope that the style of our manuscript is now more understandable.

3. We assigned the table a number. It is now referred as Table 1 in the text.

Sincerely,

Philippe Vignon, MD, PhD